Author's response to reviews

Title: Towards an understanding of medical student resilience in longitudinal integrated clerkships: a grounded theory approach

Authors:

Jennene A Greenhill (Jennene.greenhill@flinders.edu.au)
Ken R Fielke (ken.fielke@gmail.com)
Janet N Richards (Janet.richards@flinders.edu.au)
Leesa J Walker (leesa.walker@flinders.edu.au)
Lucie K Walters (Lucie.walters@flinders.edu.au)

Version: 2
Date: 30 January 2015

Author's response to reviews:

This article was initially submitted to Academic Medicine. The cover letter describes the revisions made to the original manuscript in response to Academic Medicine referees.